Author's response to reviews

Title: Malignant fibrous histiocytoma of the face: report of a case.

Authors:

Laszlo Seper (seper@uni-muenster.de)
Richard Schwab (schwab@kkk.org.hu)
Sirichai Kiattavorncharoen (kiattav@hotmail.com)
Andre Buchter (a.buechter@citykom.net)
Agnes Bankfalvi (agnes.bankfalvi@uk-essen.de)
Ulrich Joos (joos@uni-muenster.de)
Jozsef Piffko (piffko@uni-muenster.de)
Birgit Kruse-Loesler (losler@uni-munster.de)

Version: 6 Date: 8 October 2007

Author's response to reviews:

Dear Sir,

Thank you for the useful comments on our manuscript. According your suggestions the following changes are made.

1) Sentence case are used in the title
2) Each affiliation are written in full and be provided in the necessary format
3) Author email addresses were updated and give on the title page, beneath the corresponding author details
4) Underlining was changed it to bold in the abstract section
5) Case report section was renamed to Case presentation
6) Keywords section was removed from the manuscript
7) The reference list contain all authors names
8) Figure legends include a Figure legends section heading
9) Author contributions was moved to after the conclusions and before the references section
10) An Acknowledgements section was included in the manuscript directly above the References with a statement